Author’s response to reviews

Title: Limb girdle muscular dystrophy type 2G with myopathic-neurogenic motor unit potentials and a novel muscle image pattern

Authors:

Ana Cotta (cotta@sarah.br)
Julia F Paim (julia@sarah.br)
Antonio L Cunha-Junior (400564@sarah.br)
Rafael X Neto (9679@sarah.br)
Simone V Nunes (7713@sarah.br)
Monica M Navarro (400039@sarah.br)
Jaquelin Valicek (400585@sarah.br)
Elmano Carvalho (9230@sarah.br)
Lydia U Yamamoto (lydiauy@usp.br)
Camila F Almeida (camila.almeida@usp.br)
Shelida V Braz (12758@sarah.br)
Reinaldo I Takata (202911@sarah.br)
Mariz Vainzof (mvainzof@usp.br)

Version: 4 Date: 4 September 2014

Author’s response to reviews:

Dear Dr Magdalena Morawska

Executive Editor

BMC Clinical Pathology

We are very grateful for the reviewer’s positive comments about our manuscript. We have accepted and corrected all the suggestions accordingly. In addition, the EMG data for the Diagnostic Assessment section was rewritten, and a new table for these data was included (previous lines 103-109). The doubt about the pseudoneurogenic pattern was corrected also (lines 174-178). In line 175 “nerve necrosis” was replaced by “nerve degeneration” and the paragraph was ameliorated.

We are submitting the revised manuscript, with all the modifications highlighted. We hope this MS is now suitable for evaluation and publication.

With kind regards.

Mariz Vainzof